Author’s response to reviews

Title: Evaluation of peptide-rich root extracts of Calliandria portoriscensis (Jacq.) Benth (Mimosaceae) for in vitro antimicrobial activity and brine shrimp lethality

Authors:

Omonike Ogbole (nikeoa@yahoo.com)

Nkiruka Ndabai (sharedafun2000@gmail.com)

Toluwanimi Akinleye (toluwanimiemmanuel@rocketmail.com)

Alfred Attah (alfattah2000ng@gmail.com)

Version: 4 Date: 24 Jan 2020

Author’s response to reviews:

Sueli Fumie Yamada-Ogatta, Ph.D (Reviewer 1): Thanks for the replies. There are minor typing corrections that can be made during the proof step.

Query: Please type in full the following genus names: Enterococcus faecalis (instead of E. faecalis) and Candida albicans (instead of C. albicans). “Typhimurium” is a serovar of Salmonella enterica species. Thus, it should be typed in capital letter without italic: Salmonella Typhimurium.

Response: Corrections have been effected in the manuscript; the scientific names of organisms are now typed in full and italicised; Salmonella Typhimurium is typed in capital letter.

Cristiane Y. Koga-Ito, Ph.D. (Reviewer 3):

The authors revised the manuscript properly. All the raised questions were adequately answered.

Response: Thank you!